Author's response to reviews

Title: Care for post-stroke patients at Malaysian public health centres: Self-reported practices of Family Medicine Specialists

Authors:

Aznida F Abdul Aziz (aznidaf@yahoo.com)
Nor Azlin Mohd Nordin (azlin8@yahoo.com)
Noor Abd Aziz (azah@ppukm.ukm.edu.my)
Suhazeli Abdullah (suhazeli@yahoo.com)
Saperi Sulong (saperi@ppukm.ukm.edu.my)
Syed M Aljunid (saljunid@unu.edu)

Version: 3 Date: 20 February 2014

Author's response to reviews: see over
Care for post-stroke patients at Malaysian public health centres: Self-reported practices of Family Medicine Specialists
Aznida F Abdul Aziz, Nor Azlin Mohd Nordin, Noor Abd Aziz, Suhazeli Abdullah, Saperi Sulong and Syed M Aljunid

Revised Manuscript -details

<table>
<thead>
<tr>
<th>Reviewer’s Comments</th>
<th>Comments</th>
<th>Reference in manuscript</th>
</tr>
</thead>
<tbody>
<tr>
<td>There is one point that still needs to be addressed. The authors have not responded to the request by reviewer KP for an explanation in the paper why results of only two of the 7 domains are reported. You have added the knowledge on stroke management. But what about the other components?</td>
<td>The other components will be discussed in another manuscript which will address the adequacy of post stroke care provision at community level.</td>
<td>Page 9 lines 209-213</td>
</tr>
</tbody>
</table>